Author's response to reviews

**Title:** Clinical Correlates of Chronic Cerebrospinal Venous Insufficiency in Multiple Sclerosis

**Authors:**

- Bianca Weinstock-Guttman (BWeinstock-Guttman@KaleidaHealth.Org)
- Murali Ramanathan (Murali@buffalo.edu)
- Karen Marr (KMarr@BNAC.net)
- David Hojnack (DHojnack@theJNI.org)
- Ralph H.B. Benedict (Benedict@Buffalo.edu)
- Charity Morgan (cmorgan@ms.soph.uab.edu)
- Eluen A Yeh (AYeh@theJNI.org)
- Ellen Carl (ECarl@BNAC.net)
- Cheryl Kennedy (CKennedy@BNAC.net)
- Justine Reuther (JReuther@BNAC.net)
- Christina Brooks (CBrooks@BNAC.net)
- Kristin Hunt (KHunt@BNAC.net)
- Makki Elfadil (makkielfadil@hotmail.com)
- Michelle Andrews (MAndrews@BNAC.net)
- Robert Zivadinov (RZivadinov@BNAC.net)

**Version:** 3  **Date:** 26 November 2011

**Author's response to reviews:**

Please find attached a revised version of our manuscript. The revision addresses all of the Reviewers' comments. We have addressed all of the comments. The changes to the manuscript are highlighted in the manuscript and a point-by-point response in summarized in Response to Reviewers.